Title: How does community context influence coalitions in the formation stage? A multiple case study based on the Community Coalition Action Theory.

Authors:

   Michelle C Kegler (mkegler@sph.emory.edu)
   Jessica A Rigler (Rigler@azdhs.gov)
   Sally Honeycutt (shoneyc@sph.emory.edu)

Version: 5 Date: 18 February 2010

Author's response to reviews: see over
Dear Editor:

We made the following edits as requested:

Authors' contributions: Indicated that all authors read and approved the final manuscript.

Removed capitalization in the title except for the name of the theory (a proper noun)

Included the country in the affiliation details.

Removed the title and keywords from the abstract.

Changed underlining to bold.

Moved Authors' contributions directly after the competing interests section.

Let me know if you require any additional changes.

Thanks!  Michelle Kegler